Author's response to reviews

Title: 2 Novel Deletions of the Sterol 27-hydroxylase Gene in a Chinese Family with cerebrotendinous xanthomatosis

Authors:

Tian Di Jr (tandi320@163.com)
Zhang Zaiqiang Dr (zhangzq007@sohu.com)

Version: 4 Date: 22 September 2011

Author's response to reviews:

1. Superscript numbers were corrected.
2. The structure of abstract was changed.